Author's response to reviews

Title: Improving education in primary care: development of an online curriculum using the blended learning model

Authors:

  Linda Orkin Lewin (llewin@peds.umaryland.edu)
  Mamta Singh (mamta.singh@case.edu)
  Betzi L. Bateman (betzi.bateman@gmail.com)
  Pamela Bligh Glover (pamela.glover@case.edu)

Version: 7 Date: 8 June 2009

Author's response to reviews:

typographical errors
a few redundant punctuations changed
no content change was made